CMSC 828G:Spring 2002 Lise Getoor Homework 0

1. Name:

Major: Degree: E.mail:

2. Do you intend to take this class for:

Credit Audit Undecided 3. If you are a CS student, do you intend to take this class as a MS/PhD Comps class for:

AI Credit DB Credit N/A 4. Please specify database classes you have taken:

5. Please specify AI classes you have taken:

6. Have you taken:

course no yes (indicate semester and year) CMSC726 - Machine Learning CMSC724 - Database Management Systems CMSC838L - Information Retrieval Systems Statistics course Probability course 7. Please rate your knowledge of Matlab:

None Beginner Intermediate Proficient 8. Please list any topics you would like to see covered in this class.

9. Please list any topics you would prefer not to be covered in this class.

10. Additional comments: